Author's response to reviews

Title: Study protocol: Follow-up home visits with nutrition: a randomised controlled trial

Authors:

Anne Marie Beck (amab@life.ku.dk)
Stine Kjær (stinekjaer@live.dk)
Birthe S Hansen (bsha@life.ku.dk)
Rikke L Storm (rikkelunau@hotmail.com)
Kirsten Thal-Jantzen (kthaljantzen@msn.com)

Version: 3 Date: 12 November 2011

Author's response to reviews: see over
Study protocol: Follow-up home visits with nutrition: a randomised controlled trial
MS: 1837930205663199

To the Editorial Office

Thank you very much for the acceptance of our manuscript. In our revised version we have tried to follow the template for manuscripts. Also we have tried to specify the contributions of each author. Further I have a new job position, and therefore there is a change in the information about how to contact me (i.e. the corresponding author) I hope these changes are sufficient.

On behalf of the co-authors, yours sincerely
Anne Marie Beck